

## **Author's response to reviews**

**Title:** The Impact of Bone Morphology on the Outcome of the Pivot Shift Test: A Cohort Study

### **Authors:**

Thomas Branch (doctorbranch@yahoo.com)

Shaun Stinton (stinton@gmail.com)

Adrija Sharma (adrija.sharma@gmail.com)

Frederic Lavoie (fredericlavoie.md@gmail.com)

Christian Guier (guier@jhmail.com)

Philippe Neyret (philippe.neyret@chu-lyon.fr)

**Version:** 3 **Date:** 28 Aug 2017

### **Author's response to reviews:**

1. The following was added to the paper to address the ethics approval question.

Subjects consented to participate in the study. Institutional review board approval was not required for this study at the institution where testing was performed (Centre Albert-Trillat, Hôpital de la Croix-Rousse, Lyon, France) at the time of the study.

2. A supplementary files section was added at the end of the paper.